Author’s response to reviews

Title: How have researchers defined and used the concept of “continuity of care” for chronic conditions in the context of resource-constrained settings? A scoping review of existing literature and a proposed conceptual framework

Authors:

Lana Meiqari (lana.meiqari@vu.nl)
Tammam AL-OUDAT (Tammam.ALOUDAT@geneva.msf.org)
Dirk Essink (d.r.essink@vu.nl)
Fedde Scheele (f.scheele@olvg.nl)
Pamela Wright (pamelagihc@gmail.com)

Version: 1 Date: 01 Dec 2018

Author’s response to reviews:

Thank you for your comment and suggestions. As per your suggestion to strengthen the connection between CoC and health policy, paragraphs in the abstract, introduction and discussion were revised to explain the focus on CoC within the wider field of health policy and system research and practice.